Author's response to reviews

Title: Chronic hepatitis E virus infection in a leukemia patient presenting with elevated transaminases: a case report

Authors:

Annika Braun (Annika.Braun@med.uni-heidelberg.de)
Jürgen J. Wenzel (Juergen.Wenzel@klinik.uni-regensburg.de)
Christa Flechtenmacher (Christa.Flechtenmacher@med.uni-heidelberg.de)
Mojdeh Heidary Navid (Mojdeh.Heidary-Navid@med.uni-heidelberg.de)
Christoph Eisenbach (Christoph.Eisenbach@med.uni-heidelberg.de)
Wolfgang Jilg (Wolfgang.Jilg@klinik.uni-regensburg.de)
Wolfgang Stremmel (Wolfgang.Stremmel@med.uni-heidelberg.de)
Paul Schnitzler (Paul_Schnitzler@med.uni-heidelberg.de)

Version: 3 Date: 22 July 2012

Author's response to reviews: see over
Editorial office

1. An Author’s Contribution section has been added, see end of the manuscript, marked in bold:
   Author’s contribution

   AB, JJW, CF, MHN, CE, WJ and WS analyzed and interpreted the patient data regarding chronic HEV infection. JJW performed PCR analysis, AB and PS wrote the manuscript. All authors read and approved the final manuscript.

Reviewer #1

1. We fully agree with the reviewer and have corrected the conclusions in the Abstract, marked in bold: “Usually, HEV infection appears as an acute infection; rare chronic infections have only been reported for transplant patients, HIV-infected patients and patients with haematological malignancies.”

2. Quality of English has been improved, a native English speaker has revised the manuscript, see bold marked sentences in the revised manuscript.